Author's response to reviews

Title: A survey of the use of complementary medicine by two community groups of women with polycystic ovary syndrome.

Authors:

Susan Arentz Ms (s.arentz@uws.edu.au)
Caroline A Smith Professor (caroline.smith@uws.edu.au)
Jason A Abbott A/Professor (j.abbott@unsw.edu.au)
Alan Bensoussan Professor (a.bensoussan@uws.edu.au)

Version: 8 Date: 19 November 2014

Author's response to reviews: see over
BMC Alternative and Complementary Medicine,

Dear Mr Prozenko,

Please find attached our revised paper titled: ‘A survey of the use of complementary medicine by two community groups of women with polycystic ovary syndrome.’

Thank you for your advise regarding clarification of our article. We have inserted two new sentences as per your recommendations (line 223-227).

We have included the survey as an attachment.

We hope the revisions to this article are met with satisfaction. Further revision is most welcome.

Thank you again for considering our paper for publication in the BMC Alternative and Complementary Medicine.

Yours sincerely

Susan Arentz